Author's response to reviews

Title: Obesity prevalence estimates in a Canadian regional population of preschool children using variant growth references

Authors:

Laurie K Twells (ltwells@mun.ca)
Leigh Anne Newhook (lnewhook@mun.ca)

Version: 3 Date: 7 February 2011

Author's response to reviews: see over
February 4th, 2011

Dear Dr. Norton,

Thank you for accepting our paper for publication in your journal. We have made the suggested changes to the manuscript.

Under major revisions, the author list has been edited as requested including the details around affiliation. In addition, the references have been cited in the text using square brackets and numbered in consecutive order. Under minor revisions, we have removed all highlighting and tracking and reviewed the article carefully for all typographical, grammatical errors.

Please accept our newly revised manuscript for publication.

Yours sincerely,
Laurie Twells

Laurie Twells PhD
Assistant Professor
School of Pharmacy & Faculty of Medicine
Memorial University
ltwells@mun.ca